Author's response to reviews

Title: Use of a blood gas analyzer and a laboratory autoanalyzer in routine practice to measure electrolytes in intensive care unit patients

Authors:

Yasemin U Budak (yaseminbudak2000@yahoo.com)
Kagan Huysal (khuysal@yahoo.com)
Murat Polat (drmpolat@yahoo.com)

Version: 2 Date: 1 February 2012

Author's response to reviews:

We get institutional ethics committee approval.